Author’s response to reviews

Title: Service use, unmet needs, and barriers to services among adolescents and young adults with autism spectrum disorder in Poland

Authors:

Mateusz Platos (mateusz.platos@gmail.com)

Ewa Pisula (ewa.pisula@psych.uw.edu.pl)

Version: 2 Date: 06 Aug 2019

Author’s response to reviews:

Dear Editor Ehsan Zarei,

Thank you very much for the opportunity to revise our paper. We have prepared a final, clean version of the manuscript and changed the format of the references to Vancouver style. We have also made technical changes regarding authors’ email addresses and the order of the declarations.

Kind regards,

Mateusz Platos